Author's response to reviews

Title: Exosome derived from epigallocatechin gallate treated breast cancer cells suppresses tumor growth by inhibiting tumor-associated macrophage infiltration and M2 polarization

Authors:

Ji-Young Jang Dr. (jjy93@hanmail.net)
Jong-Kuen Lee Mr. (fz2jk@naver.com)
Yoon-Kyung Jeon Prof. (junarplus@chol.com)
Chul-Woo Kim Prof. (cwkim@snu.ac.kr)

Version: 7 Date: 27 August 2013

Author's response to reviews: see over
Please include the full name of the animal ethics committee that approved the study

Page 10

Animals experiment for this research was designed and carried out according to the standard guideline of Institutional Animal Care and Use Committee (IACUC), and the study design had been approved by Seoul National University Institutional Animal Care and Use Committee (http://iacuc.snu.ac.kr).